Author's response to reviews

Title: Posttraumatic stress symptoms and health-related quality of life: a two year follow-up study of injury treated at the Emergency Department

Authors:

Juanita A Haagsma (j.haagsma@erasmusmc.nl)
Suzanne Polinder (s.polinder@erasmusmc.nl)
Miranda Olff (m.olff@amc.uva.nl)
Hidde Toet (h.toet@veiligheid.nl)
Gouke J Bonsel (g.bonsel@erasmusmc.nl)
Ed F van Beeck (e.vanbeeck@erasmusmc.nl)

Version: 6 Date: 4 January 2012

Author's response to reviews:

January 4th, 2012

Concerning – resubmission manuscript 2105678078526649

Posttraumatic stress symptoms and health-related quality of life: a two year follow-up study of injury treated at the Emergency Department

Dear members if the BMC Series Editorial Production Team,

We revised the manuscript as recommended by The BMC Series Editorial Production Team dd 30thst December 2011.

We have added citation of figure 1 in the last sentence of the paragraph “Association of posttraumatic stress symptoms indicative of PTSD (IES#35) with HRQoL” of the results

The article is revised on the BMC Psychiatry web page. In case of further queries we are most willing to help.

Kind regards,

Juanita Haagsma
ErasmusMC
Department of Public Health
Rotterdam, The Netherlands